Author’s response to reviews

Title: An example of dark side of the liver diseases; PORPHYRIA CUTANEA TARDA: a case report

Authors:

Hanife Usta Atmaca (hanifeusta@yahoo.com)
Feray Akbas (atlibatur@yahoo.com)

Version: 2 Date: 12 Dec 2018

Author’s response to reviews:

Dear reviewers,

Thank you very much for the kind effort to review our manuscript and for the detailed guidance to improve our paper.

Our answers to reviewers’ comments are as follows:

Reviewer #2: All text is checked for spelling mistakes and corrected.

First sentence is rewritten as ‘Porphyrias are a rare group of metabolic disorders that can either be inherited or acquired’

Reviewer #3: 1. A paragraph is added to conclusion explaining what is unique about the case in our opinion.

2. CARE guidelines are checked and followed and the title is re-written including the words "case report".

3. English is revised, all text is checked for spelling and punctuation errors. Discussion is re-organised to allow a better understanding of the text.

Regards,

H.U. Atmaca and F. Akbas